

EO-1 Weekly Report


	Earth Observing – 1

	August 15, 2012 – August 21, 2012

	DOY
	228
	–
	234

	Mission Day
	4346
	–
	4352


EO-1 SPACECRAFT

Instruments

· Scheduled 139 science Data Collection Events (DCEs) this week 

· Received all images
· Performed Instrument calibration

· ALI Internal Calibration I 233/00:29z

EO-1 MISSION OPERATIONS CENTER

Real-Time

· Maintaining nominal spacecraft state of health
· Conducted a Contingency Plan Exercise to verify failover capibilities

· Conducted a pass with WGS that was sent from the ground station to the Backup FEDS system and manned using the ASIST 3rd string T&C system.

· Conducted a TDRS pass through the backup FEDS system and manned using the ASIST 3rd string T&C system.

· Both passes went nominally, with data received, commands sent and a data dump performed on the GS pass.

· Monitoring close approach of object 32454 (Fengyun 1C Debris) on day 239
· Object appears to have moved out of range, but keeping an eye on it
· The first X-Band test with TrollSat contained an incorrect station identifier (JPL ASPEN used MGS as the target instead of TR2). 

· The S-band went fine but the images were not transmitted to TR2. 

· The change has been made at JPL to fix this problem and a new X-Band test is scheduled for next week.
· The Single, Slow scan Lunar Calibration had the full moon with no clipping.
· The sample rate of the previous scans was about 28x, we are reducing the speed of the scan to increase the sample size to about 35x.

· Looking into timing reductions to make the image size smaller (less empty space) which would enable the use of ALI on future scans.

· Creating scripts to ease operations and reduce possibilities for errors

· Creating scripts to switch from the primary FEDS to the backup FEDS to ease the transition during a failover.

· Testing scripts to clean up log files and directories.

· Testing monthly cleanup script that will call the individual cleanup scripts

· Testing scripts to transfer primary ASIST T&C and update the central file hub that we are building.

· Adding functionality to prepare the target ASIST system for primary use automatically

· Updating the RTS / TSM binder to show the most recent onboard RTS and TSMs.

· Researching all files received on ASIST systems to see if they can all be sent to the central file hub and then transferred all Realtime systems
Mission Planning
· Working with White Sands and the TR2 personnel to schedule some TR2 X-Band Engineering Passes

· Working with JPL to configure Lunar Calibration sequences 
· Looking over the UDAP file provided by JPL for the single scan (HSI, ALI, BOTH) to determine if all necessary commands are there and the timing is accurate

· Ability to do multiple Hyperion scenes in one lunar cal needs multiple base file ID numbers
· Researching updates on MP files to determine if they can be gathered on the central file hub and then transferred to both MP systems
· Rewriting some SOPs for mission planning
Flight Dynamics
· Looked at the possible conjunction listed above, input into STK to show geometry of the collision
· Working with ASPEN technicians to rebuild lunar calibration

· Experimenting with various lunar calibration timings and instrument configuration

· Researching updates to FD files to determine if they can be gathered on the central file hub and then transferred to both FD systems
· Rewriting some SOPs for flight dynamics
System Administration

· Submitted updates for two Open IONet firewall (OGRS) rulesets. The rules allowing the EO-1 FD and DPS systems to send products to a legacy EDC computer were removed as this system is no longer used operationally by EDC.

· Continued effort to automate system backups based on pass schedule. The scripts appear to be complete, but additional testing is required before using them operationally.

· Completed the script that reads/parses the daily file to determine sizable gaps in the pass schedule. At jobs are created during the gaps to run the backup manager script.

· Completed the backup manager script that attempts to backup as many systems as possible in the gap before the next AOS time. There must be at least 30 minutes left in order for another backup to be started.

· Configured certificate based SSH between the EO-1 backup server and the non-privileged SA account on each system it will be backing up.

· Added sudo entries on each Linux system that will allow the non-privileged SA account to run the backup script.

· Working with EDOS personnel to configure access to the DPS systems for X-band data delivery. Firewall rules are already in place. Will need to establish certificate based SFTP for their automated delivery scripts.

· Completed exercises for the EO-1 contingency plan test. Worked with FOT to schedule two passes (Wallops and TDRS) to connect to our backup FEDS computer. A backup ASIST computer was also used to take these passes. Will need to work with IT personnel to complete the write-up associated with the contingency plan exercise.

· Completed applying the Aug 2012 updates/patches on the IONet Windows computers. These patches included Adobe Flash, Adobe Reader, Firefox, and Microsoft products.

· Completed applying the Aug 2012 updates/patches on the CNE computers. The patches included Adobe Flash, Adobe Reader, Firefox, and Microsoft products.

· Assisted FOT with routine password changes for operational computers in the MOC.

· Created backups for the ASIST computers, flight dynamics computers, ASPEN computers, countdown clock, data processing computers, FEDS computers, firewall, trending system, and MOC CNE computer.

· Created weekly tape.

GROUND AND SPACE NETWORK

Station Downtimes 

· SGS has had an equipment malfunction and latest indication is that it will be back up August 2012
· HGS is down, unknown time to be back up

Operational Discrepancies

· None
Number of Real-Time Supports

[image: image1.emf]Date DOY S/X-Band S-Band Only Non-DSMC TDRSS Daily Totals

15-Aug-12 228 4 4 1 0 9

16-Aug-12 229 6 4 1 0 11

17-Aug-12 230 5 3 0 0 8

18-Aug-12 231 5 4 0 0 9

19-Aug-12 232 6 4 1 0 11

20-Aug-12 233 5 6 1 0 12

21-Aug-12 234 5 3 0 1 9

36 28 4 1 69 Weekly Totals


UPCOMING EVENTS

· Slowing scan of HSI Only Lunar Calibration

· Redesign of nominal Lunar Calibration
· TR2 X-Band pass testing, s-band troubleshooting
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		Date		DOY		S/X-Band		S-Band Only		Non-DSMC		TDRSS		Daily Totals

		15-Aug-12		228		4		4		1		0		9

		16-Aug-12		229		6		4		1		0		11

		17-Aug-12		230		5		3		0		0		8

		18-Aug-12		231		5		4		0		0		9

		19-Aug-12		232		6		4		1		0		11

		20-Aug-12		233		5		6		1		0		12

		21-Aug-12		234		5		3		0		1		9

		Weekly Totals				36		28		4		1		69
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